

Format for ANSWERING REVIEWERS



March 12, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 8713-review.doc).

Title: Enhanced microfracture techniques in cartilage knee surgery: fact or fiction?

Author: Dr. med. Stefan Bark

Name of Journal: *World Journal of Orthopedics*

ESPS Manuscript NO: 8713

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) 02446114

(2) 2565607

(3) 467030

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Orthopedics*.

Sincerely yours,

A handwritten signature in black ink that reads 'Stefan Bark'.

Stefan Bark, M.D.

University of Schleswig-Holstein

Department of Trauma and Orthopaedics

Ratzeburger Allee 160

23538 Luebeck

Germany

+49 451 500 2642

stefan.bark@uksh.de